\begin{table}%t1
\caption{ General properties of the nearby galaxies with radio SNRs.\label{tab1}}
%\centerline
{\begin{tabular} {llllll}  \hline\hline \noalign{\smallskip}
Galaxy & Hubble &  Major and minor & Distance &
Incl. angle & Number of \\
& type$^{\dagger}$ & (arcmin)$^{\dagger}$ & (Mpc) &
(degrees)$^{\ddagger}$
& radio SNRs \\
\noalign{\smallskip} \hline
LMC& SB(s)m & $645\times550$ &$0.055^{a}$ & \hfil 35 &\hfil 52\\
SMC& SB(s)m pec & $320\times185$ &$0.065^{a}$&\hfil 61 &\hfil 12\\
M~31      & SA(s)b & 190 $\times$ 60 & $0.75^{b}$ & \hfil 78 &\hfil 30 \\
M~33      & SA(s)cd & 70.8 $\times$ 41.7 & $0.82^{b}$ &\hfil 56 &\hfil 53\\
IC 1613  & IB(s)m & 16.2 $\times$ 14.5 & $0.69^{b}$ &\hfil 27 &\hfil 1\\
NGC 300  & SA(s)d & 21.9 $\times$ 15.5 & $2.1^{c}$ &\hfil 46 &\hfil 17\\
NGC 6946 & SAB(rs)cd & 11.5 $\times$ 9.8 & $5.1^{d}$ &\hfil 42 &\hfil 35\\
NGC 7793 & SA(s)d & 9.3 $\times$ 6.3 & $3.38^{e}$ &\hfil 50 &\hfil 7 \\
M~82 & I0 & 11.2 $\times$ 4.3 & $3.9^{f}$ &\hfil 66 &\hfil 50 \\
NGC 1569 & IBm & 3.6 $\times$ 1.8 & $2.2^{g}$ &\hfil 64 &\hfil 3 \\
NGC 2146 & SB(s)ab pec & 6.0 $\times$ 3.4 & $14.5^{h}$ &\hfil 36 &\hfil 3 \\
\noalign{\smallskip} \hline
\end{tabular}}
\par
\smallskip
Note: $^{\dagger}${NED Database}; $^{\ddagger}${Tully (\cite{Tully88})};
$^{a}${Filipovi\' c (\cite{Filipovic02})}; $^{b}${Freedman et~al. (\cite{Freedman01})};
$^{c}${Freedman et~al. (1992)};
$^{d}${de Vaucouleurs (\cite{deVaucouleurs79})}; 
$^{e}${Puche \& Carignan (\cite{Puche88})};
$^{f}${Sakai \& Madore (\cite{Sakai99})}; 
$^{g}${Israel (\cite{Israel88})};
$^{h}${Tarchi et~al. (\cite{Tarchi00})}.
\end{table}